Author's response to reviews

Title: Cardiovascular Risk Factors among Chamorros

Authors:

Binh Chiem (bchiem@ucsd.edu)
Victoria Nguyen (v19nguye@ucsd.edu)
Phillis L Wu (plwu@ucsd.edu)
Celine M Ko (cmko@ucsd.edu)
Lee Ann Cruz (lcruz@san.rr.com)
Georgia R Sadler (gsadler@ucsd.edu)

Version: 6 Date: 5 December 2006

Author's response to reviews:

BioMed Central Editorial Production Team,

Thank you for pointing out the changes that need to be made in this manuscript. The following changes have been implemented:
1. The title has been changed so all unnecessary capitals have been removed;
2. The list of authors in the manuscript have been written identically and in the order to match what is listed in the submission system;
3. The authors' qualifications have been removed;
4. Affiliations have been written in full and now include department, institute, city, and country;
5. The word count has been removed from the manuscript;
6. The repeated title after the title page has been removed;
7. The section headings have been changed and all unnecessary capitals have been removed;
8. The keywords section has been removed;
9. The introduction section has been renamed 'Background';
10. The references now include all authors' names and the term 'et al.' has been removed;
11. The authors declare no competing interests;
12. All of the authors' read and approved the final manuscript.

Please contact me if you have any further questions.

Georgia Robins Sadler